\begin{table}%T3
\caption{Properties of previously identified CSS sources and candidates.
Parameters are the same as for Tables~\ref{tab1} and~\ref{tab2}.
}
\label{tab3}
%%\centerline
\par
%\begin{center}
\par
{
\begin{tabular}{l l l c c c c rl c c}
\hline\hline
\noalign{\smallskip}
Source & B1950 & Refs.~$^{{a}}$ & Type & $z$ & $S_{4.8}$ & $m_{4.8}$ & \multicolumn{2}{c}{$p_{4.8}$} & $\alpha$ & Flag \\
 &  & &  &  & (Jy) &  & \multicolumn{2}{c}{(\%)} &  &  \\
\noalign{\smallskip}
\hline
\noalign{\smallskip}
J0405$-$1308& 0403$-$132& K97 &Q & 0.571& 2.30& 0.05&   2.40   &     & $-$0.79& e \\
J0627$-$0553& 0624$-$058& K97 &EF&  ... & 4.92& 0.03&   6.53   &     & $-$1.15& e \\
J1018$-$3144& 1015$-$314& K97 &Q & 1.246& 1.41& 0.02&$<$0.5~~~&(0.1)& $-$0.92& c \\
J1150$-$0023& 1148$-$001& K97 &Q & 1.983& 1.60& 0.03&   2.88   &     & $-$0.53& l \\ 
J1218$-$4600& 1215$-$457& K97 &Q & 0.529& 2.08& 0.03&   2.23   &     & $-$0.88& e \\
J1242$-$0446& 1239$-$044& M92 &G & 0.480& 0.61&  ...&   ....   &     & $-$1.20& e \\
J1546$+$0026& 1543$+$005& G83 &G & 0.550& 0.94& 0.05&$<$0.5~~~&     & $-$0.59& l \\
J1556$-$7914& 1549$-$790& K97 &G & 0.150& 3.73& 0.02&$<$0.5~~~&     & $-$0.74& c \\
\hline
\end{tabular}}
\\[1mm]
%\medskip
\parbox{13.5cm}{
$^{{a}}$ References:
G83 = \cite{gop83}; 
K97 = \cite{kin97};
M92 = \cite{man92}.}
%\end{center}
 \end{table}